Author's response to reviews

Title: Mortality among adults: gender and socioeconomic differences in a Brazilian city

Authors:

Ana Paula Belon (paulabelon@gmail.com)
Marilisa BA Barros (marilisa@unicamp.br)
Letícia Marín-León (leticia@fcm.unicamp.br)

Version: 6 Date: 16 January 2012

Author's response to reviews: see over
Cover letter

Dear Editors,

The BMC Series Editorial Production Team only requested the following formatting change:

References: reference entry [41] is incomplete. Please supply the volume number and page range of this reference.

Unfortunately, as this article is in press, we cannot provide the volume number and page range. For the reference #41 (below), we followed the reference style for In press article, which is described in the Examples of the BMC Public Health reference style at webpage http://www.biomedcentral.com/bmcpublichealth/authors/instructions/researcharticle#reference-style. The only additional information in this reference was the year “2012”. We can assure that the publication will be launched in this year, 2012, in the volume 28.


We are grateful to the Editorial Production Team for the formatting work. We point out some problems below and hope that they can be solved.

- In the Table 3, please change capital letter B in the 3\textsuperscript{th} column (“Middle”) for b (lowercase)

- In the reference number 22, the correct title is “Projeções populacionais para pequenas áreas: métodos e aplicações”.

- At the first page, we were wondering if you could replace the name of our department, which was recently changed from “Department of Preventive and Social Medicine” to “Department of Collective Health”. Then, the correct would be:
  - Department of Collective Health, School of Medical Sciences, State University of Campinas, São Paulo, Brazil.

- Please, between Abstract and Background, could you insert the following keywords?
o Health Inequalities;
 o Social Inequalities;
 o Socioeconomic status;
 o Mortality;
 o Causes of Death;
 o Gender;
 o Brazil

Sincerely,

The authors